DIRECTED GRAPH TRANSFORMERS MEET METABOLIC NETWORKS

Anonymous authors

003 004

006

008 009

010 011

012

013

014

015

016

017

018

019

021

023

024

025

026 027 028

029

Paper under double-blind review

ABSTRACT

Technical advances in sequencing have allowed the reconstruction of genome-scale metabolic models (GEMs) for a wide range of microorganisms. These models have been particularly useful for the prediction of essential genes and reactions, which are potential targets for antimicrobial therapies. However, current methods for essentiality prediction are computationally limited and are not able to accommodate the increasingly available data. Motivated by the success of data-driven approaches in other domains, this work introduces the metabolic transformer, a model designed for holistic identification of essential reactions in genome-scale models, entirely trained on synthetic knock-out data. It is demonstrated that the problem of essential reaction prediction can be theoretically formulated as the identification of redundant nodes in directed bipartite graphs. This reveals the limitations of message-passing schemes and motivates the development of a novel graph transformer architecture specifically tailored for metabolic networks. The proposed architecture is capable of addressing the essential reaction identification problem by capturing both the directionality and global structure of metabolic networks. To demonstrate the effectiveness of our approach, we composed a large-scale dataset of genome-scale models reconstructed from real microorganisms.¹.

1 INTRODUCTION

In recent years, the emergence of high-throughput technologies allowed the integration of transcriptomic data of multiple pathogens into large biological datasets. This integration paved the way for the reconstruction of microorganism metabolism which directly led to the possibility of modelling these systems computationally Ric (2020).

Metabolism is the set of basic life processes that take place in the cell. All the metabolic chemical reactions that occur in a cell form a metabolic network. Genome-Scale models (GEM) are structured biochemical, genetic, and genomic databases for an organism, that aim to cover the whole metabolic network of a cell. As of 2019, more than 6.000 GEMs have been reconstructed for organisms including bacteria, archaea and eukaryotes and over 140.000 automated reconstructions are available from over 2.600 organisms Büchel et al. (2013); Gu et al. (2019); Monk et al. (2017).

GEMs have been particularly useful for phenotype simulation and prediction. These models have
 shown great accuracy in predicting the growth rate of microorganisms along with the flux carried
 out for each reaction of metabolism Orth et al. (2010). This has been particularly interesting for the
 identification of essential reactions and genes. Essential reactions are those that are required for the
 growth of the organism Henry et al. (2006). Consequently, when these reactions are knocked out
 the growth in the model is null, i.e., the organism dies. The genes that encode these reactions are
 appealing therapeutical targets.

Although the results on essentiality prediction are promising, current methods can be computationally demanding and their application to uncertain data can be challenging. Improving the accuracy of predictions is not trivial and requires an increasing complexity on the models that sometimes involves an exponential combinatorial cost or requires the estimation of missing data O'brien et al. (2013);
Salvy & Hatzimanikatis (2020). In addition, recent advances in metabolism profiling technology Judge et al. (2019) will produce large amounts of data that could be hard to accommodate into current

¹The dataset and source code will be made publicly available in the final version.



Figure 1: The metabolic transformer. *a*) The model is trained entirely from synthetic knock-out data obtained from metabolic network reconstructions from different microorganisms. *b*) Once trained, the model enables the computation of essential reactions in models where synthetic knock-out computations were previously infeasible. *c*) The model trained on syntehtic data can serve as a backbone for finetuning with real data.

modelling algorithms. In this context, there is a growing belief that data-driven methods (i.e. deep learning) have the potential to improve metabolic modelling Zampieri et al. (2019).

In this work, we propose a supervised learning approach, fully trained on synthetic knock-out data, 074 for the holistic prediction of essential reactions in genome-scale models using graph neural networks 075 and transformers models. Graph Neural Networks (GNN), first proposed in Scarselli et al. (2008), 076 have become the reference algorithm for learning on graph-structured data. It is well known, however, 077 that GNNs suffer from over-squashing and fail to model long-range relationships in graphs, which is a fundamental requirement for metabolic modelling. Motivated by the success of transformers 079 models in natural language processing and computer vision, many works have proposed the use of transformers for graph-structured data Min et al. (2022); Rampášek et al. (2022); Bo et al. (2023). 081 One of the early applications of transformers to graphs was presented in Dwivedi & Bresson (2020), 082 where the authors proposed a generalization of transformers for graphs by applying the attention 083 mechanism to the neighbourhood of each node. Later examples such as Kreuzer et al. (2021) and 084 Kim et al. (2022) propose approaches where attention is applied to the whole graph. Other works have also proposed the integration transformers with GNNs to capture both the local and global 085 information of the graph Rong et al. (2020); Rampášek et al. (2022).

087 This work proposes the metabolic transformer, a graph transformer model that combines directed 088 message passing with positional encodings for essential reaction prediction in genome-scale models. Previous models have been proposed before for essentiality computation on GEMs Hasibi et al. (2024). These methods however rely only on message-passing schemes or are narrowed down to a specific 090 dataset, which makes it application to other tasks or microorganisms limited. In this work, a model is 091 proposed that is able to generalize essentiality reaction prediction across different microorganisms by 092 pre-training on synthetic knock-out data. It is shown that the model can serve for effective transfer learning to real metabolic data, in this case, gene essentiality obtained experimentally in the lab. This 094 is in contrast with existing LP methods that cannot generalize outside the modelled problem. To 095 the best of our knowledge, this is the first model to successfully demonstrate transfer learning on 096 metabolic data. 097

098 099

100

102

103

105

106

065

066

067

068

069

In summary, our proposed approach has the following properties:

- For inference, the proposed method only takes as input the topology of the metabolic network. This is in contrast with previous methods that require precomputing synthetic knock-out data for inference Freischem et al. (2022); Hasibi et al. (2024).
- The method predicts the essentiality of all reactions considering the whole network at once. This property is a direct consequence of the graph transformer architecture, which can take the features of all nodes as input. This overcomes the need for GNN-based approaches that require pre-computing global features on the network Yang et al. (2022).
- Given the high throughput of deep learning models, we are able to compute predictions for all reactions from only one forward pass of the model. This is an improvement compared

with the traditional synthetic knock-out simulation, which requires the computation of the growth rate for each reaction individually Oyelade et al. (2018).

• Metabolic data available for many microorganisms is scarce or limited. Aditionally, 111 metabolic modelling is not limited to essentiality prediction, but also includes a wide 112 range of other modelling tasks such as flux prediction, control, or drug target identification. 113 By training a model on synthetic data from many diverse microorganisms, we developed a 114 model that can serve as backbone for effective transfer learning on different organisms or 115 tasks, where data can be limited. We thus believe that this is a significant step towards the 116 creation of foundational models of metabolism.

117 Currently, the majority of GEMs publicly available are built though automated reconstruction 118 pipelines Gu et al. (2019). As a consequence, most of these models lack manual curation and 119 and, hence, traditional methods for knock-out simulations based on linear programming cannot be 120 applied. In this context, since the proposed method only depends on the topology of the network, 121 and does not require any additional data to be curated, it has the potential to enable the prediction 122 of essential reactions in a wide range of models for which current methods are unable to produce 123 predictions.

124 In addition to the proposed methods, we are releasing a carefully curated dataset of 23k GEMs from 125 more than 100 different microorganisms. The dataset consists of large-scale graphs with an average 126 of 2k nodes. This is the first dataset of GEMs and, given its dimensions, we believe it is of interest 127 for the graph-learning community, making it one of the largest datasets in terms of number of nodes 128 Hu et al. (2021).

129 Although this work is particularly focused on genome-scale metabolic models, the architecture and 130 methods proposed in this work are not limited and can be applied to other topics traditionally modelled 131 with directed bipartite graphs, such as control and distributed systems Murata (1989); Heiner et al. 132 (2008), chemical reaction networks Wen et al. (2023) or retrosynthesis Chen et al. (2020). 133

The rest of the paper is organised as follows. In Section 2 preliminary definitions related to constraint-134 based models are introduced. Section 3 introduces the background for graph transformers used in 135 this work. Section 4 describes the proposed architecture, and Section 5 describes the results obtained 136 in the created dataset. Finally, Section 6 concludes the paper. 137

138 139

140

108

110

2 **PRELIMINARY DEFINITIONS**

141 In this Section, we introduce the preliminary definitions that are used throughout the paper. In 142 particular, we introduce the formal definition of constraint-based models, flux balance analysis, and essential reactions. 143

144 145

2.1**CONSTRAINT-BASED MODELS**

A constraint-based model Varma & Palsson (1994); Orth et al. (2011) is a tuple $\{\mathcal{R}, \mathcal{M}, \mathcal{S}, L, U\}$ where \mathcal{R} is a set of *reactions*, \mathcal{M} is a set of *metabolites*, $\mathcal{S} \in \mathbb{R}^{|\mathcal{M}| \times |\mathcal{R}|}$ is the stoichiometric matrix, and $L, U \in \mathbb{R}^{|\mathcal{R}|}$ are lower and upper flux bounds of the reactions.

155

146

147

148

151
152
153
154

$$2H_2 + O_2 \rightarrow 2H_2O$$
 O_2
 H_2C
 O_2
 H_2C
 O_2
 H_2C
 O_2
 H_2C
 O_2
 H_2C
 O_2
 O_2

(a) Chemical reaction.

(b) Petri net.

 H_2O

 H_{2}

156 Figure 2: A constraint-based model consist-157 ing only of the reaction in (a), can be mod-158 elled by the Petri net of (b). 159

All the reactions of the model are associated with a set of reactant metabolites and a set of product metabolites. For example, the reaction $r:A \rightarrow 2B$ has a reactant metabolite A, and a product metabolite B, with stoichiometric weights 1 and 2 respectively, i.e. reaction r consumes one molecule of type A and produces two molecules of type B. The stoichiometric matrix S accounts for all the stoichiometric weights of the reactions, i.e. S[m, r] is the stoichiometric weight of metabolite $m \in \mathcal{M}$ for reaction $r \in \mathcal{R}$.

Constraint-based models are inherently bipartite directed graphs and thus they can be represented 160 graphically as Petri nets Murata (1989); Heiner et al. (2008), where places, drawn as circles, model 161 metabolites, and transitions, drawn as squares, model reactions. The presence of an arc from a place(transition) to a transition(place) means that the place is a reactant(product) of the reaction
 modelled by the transition. The weights of the arcs of the Petri net account for the stoichiometry of
 the constraint-based model.

Example 2.1. The Petri net in Figure 2b represents a constraint-based model consisting only of one reaction, the well-known reaction of equation 2a.

168 2.2 FLUX BALANCE ANALYSIS169

Flux Balance Analysis (FBA) Orth et al. (2010) is a mathematical procedure for the estimation of steady-state fluxes in constraint-based models. FBA is generally used to predict the maximum growth rate of an organism. Let $v \in \mathbb{R}^{|\mathcal{R}|}$ be the vector of fluxes of reactions and v[r] denote the flux of reaction r. At a steady state, it holds that $S \cdot v = 0$, where S is the stoichiometric matrix. Let r_g be the reaction that models growth (or biomass production). Without loss of generality, it will be assumed that $L[r_g] \ge 0$. A theoretical optimum for the growth rate can be obtained by the linear programming problem (LPP) for FBA:

 $\begin{array}{l}
\max v[r_g] \\
\text{178} \\
st. \ \mathcal{S} \cdot v = 0
\end{array} \tag{1}$

$$L \le v \le U$$

where the maximum $v[r_g]$ obtained by the above LPP (1) is the maximum growth rate.

182 2.3 ESSENTIAL REACTIONS183

179

A reaction is said to be essential if it is required by the organism to grow. In other words, the deletion
of an essential reaction implies null growth. Consequently, these reactions have the potential to cause
the death of the modelled organism Oyelade et al. (2018).

Definition 2.1. Oyelade et al. (2018) A reaction $r \in \mathcal{R}$ is an *essential reaction* if the solution of the following LPP:

$$\max v[r_g]$$

$$st. \ \mathcal{S} \cdot v = 0$$

$$L \le v \le U$$

$$v[r] = 0$$
(2)

is equal to 0 or the LPP is infeasible. In other words, a reaction $r \in \mathcal{R}$ is essential if the maximum growth rate that the model can achieve is 0 (i.e. max $v[r_g] = 0$) when the reaction r is removed (i.e. when v[r] = 0).

The set of essential reactions, can be computed straightforwardly by solving equation 2 for each $r \in \mathcal{R}$.

Essential reactions can also be interpreted as those reactions that are required for the production of
 biomass, and that lack any alternative pathway with similiar functionality. To exemplify how the
 topology of the network conditions the essentiality of reactions, here we provide three samples of
 non-essential reactions:

Example 2.2. Let us consider the constraint-based model in Figure 3a, where reaction r_g is the biomass reaction. It can be seen that metabolite m_2 is required for the production of biomass. This metabolite can be both produced through reactions r_1 and r_2 . If reaction r_1 is removed, then reaction r_2 can still produce metabolite m_2 . Therefore, reaction r_1 is not essential. Similarly, reaction r_2 is not essential either.

Example 2.3. Let us now consider the constraint-based model in Figure 3b. Metabolite m_4 is a product of the biomass reaction. Once this metabolite is produced it requires to be evacuated from the network. This can be done through reaction r_7 or reactions r_5 and r_6 . If reaction r_7 is removed, then reactions r_5 and r_6 can still evacuate metabolite m_4 . Therefore, reaction r_7 is not essential. Similarly, reactions r_5 and r_6 are not essential either.

Example 2.4. Let us now consider the constraint-based model in Figure 3c. Here, metabolite m_3 can be diverted through the biomass reaction or through an alternative path (reactions r_8, r_9) that does not contribute to biomass production. If one of these two reactions is removed, then the other reactions can still produce metabolite m_3 . Therefore, reactions r_8 and r_9 are not essential.



Figure 3: Three sample conditions of reactions essentiality.

3 DIRECTED GRAPH TRANSFORMERS

In this section, we introduce all the concepts that compose a directed graph transformer model. We start by introducing the concept of graph neural networks, and directed graph neural networks and we end by introducing positional encodings for graphs. A definition of transformers models is provided in Appendix A.

3.1 GRAPH NEURAL NETWORKS

Graph Neural Networks (GNN) Scarselli et al. (2008); Li et al. (2015); Kipf & Welling (2016) are machine learning models that learn on data that is accompanied by a graph structure. GNNs are composed of layers of message passing networks. In each layer, the embedding vector h_i of node *i* is computed from the aggregation of the embeddings of their neighbour nodes $\mathcal{N}(i)$ of the previous layer. The initial embedding vector is usually the input feature vector that each node is given. As described in You et al. (2020), a general *k*-th GNN layer can be defined formally as:

$$h_{v}^{(k+1)} = \operatorname{AGG}\left(\left\{\operatorname{ACT}\left(\mathbf{W}^{(k)}h_{u}^{(k)} + b^{(k)}\right), u \in \mathcal{N}(v)\right\}\right)$$
(3)

where $h_v^{(k)}$ is the k-th layer embedding of node v, $\mathbf{W}^{(k)}$ and $b^{(k)}$ are the trainable weight matrix and bias respectively, ACT is an activation function and AGG is a commutative aggregation function such as maximisation, summation or mean. The different variations proposed to GNNs have shown to be very effective in learning on graphs data Hamilton et al. (2017); Veličković et al. (2017); Li et al. (2015). However, GNNs are known to suffer from limited expressivity Xu et al. (2018a), over-smoothing, this is, converging to a single solution after many layers Li et al. (2018), and oversquashing, this is, losing information from long-range nodes due to bottlenecks in the graph Alon & Yahav (2020). Their application to modelling long-range dependencies is thus considered limited.

257 258 259

229 230 231

232 233

234

235

236

237 238 239

240

247

248 249

250

251

252

253

254

255

256

3.2 DIRECTED GRAPH NEURAL NETWORKS

260 Most current GNN applications assume that graphs are undirected. However, the application of GNNs 261 on directed graphs is not straightforward. This usually requires either transforming the directed graph 262 into an undirected graph or propagating messages only over incoming (or outgoing) edges, which 263 could lead to information loss. These approaches have had a great performance on benchmarks that 264 have been historically homophilic, this is, where neighbours nodes tend to share the same label. 265 However, the case of directed and heterophilic graphs has been less explored. To overcome this, 266 in Rossi et al. (2023) the authors introduce the framework of Directed Graph Neural Networks 267 (Dir-GNN). Dir-GNNs are generic GNNs that can be applied to directed graphs by aggregating messages from both the incoming and outgoing edges of each node. This extension of GNNs showed 268 a great improvement in heterophilic datasets and seemed to outperform previous approaches for 269 directed graphs. For this reason, we will use Dir-GNNs as the base model for our models. Formally,



(a) Second eigenvector v_1 of the graph from Figure 3. (b) First element of the eigenvectors of the magnetic Network direction has been omitted to represent that Laplacian of the graph from Figure 3. The size of the directionality is missed in the combinatorial Laplacian. nodes is proportional to the real part of the eigenvector and the color is proportional to the imaginary part.

Figure 4: Unipartite representation of the graph in Figure 3 showing the spectral information of the combinatorial and magnetic Laplacians. Here values have been normalized for visualization purposes.

the k-th layer can be defined as:

281

282

283 284

285

286

287 288 289

292

293

295

$$\begin{split} m_{u,\leftarrow}^{(k+1)} &= \operatorname{AGG}\left(\left\{\operatorname{ACT}\left(\mathbf{W}_{\leftarrow}^{(k)}h_{v}^{(k)} + b_{\leftarrow}^{(k)}\right), (v,u) \in \mathcal{E}\right\}\right)\\ m_{u,\rightarrow}^{(k+1)} &= \operatorname{AGG}\left(\left\{\operatorname{ACT}\left(\mathbf{W}_{\rightarrow}^{(k)}h_{v}^{(k)} + b_{\rightarrow}^{(k)}\right), (u,v) \in \mathcal{E}\right\}\right)\\ h_{u}^{(k+1)} &= \operatorname{COM}^{(k)}\left(\left\{m_{u,\leftarrow}^{(k+1)}, m_{u,\rightarrow}^{(k+1)}\right\}\right) \end{split}$$
(4)

where $\mathbf{W}_{\leftarrow}^{(k)}$ and $\mathbf{W}_{\rightarrow}^{(k)}$ are the learnable weight matrices of the k-th layer, $b_{\leftarrow}^{(k)}$ and $b_{\rightarrow}^{(k)}$ are the learnable bias vectors of the k-th layer, ACT is an activation function, AGG is a commutative aggregation function, and $COM^{(k)}$ is a function that combines the incoming and outgoing messages. In this work $COM^{(k)}$ is implemented as: $\alpha^{(k)}m_{u,\leftarrow}^{(k+1)} + (1 - \alpha^{(k)})m_{u,\rightarrow}^{(k+1)}$ with $\alpha^{(k)} \in [0, 1]$ being a learnable parameter of the k-th layer.

302 303

3.3 POSITIONAL ENCODINGS

The attention mechanism of transformers is known to be permutation invariant. Hence, in order to capture positional information, Positional Encodings (PEs) are used. In Vaswani et al. (2017), the authors propose to add a sinusoidal PE to capture the position of each word in the sentence. However, the generalization of the sinusoidal signal to graph-structured data is not trivial. The most common approach for computing positional encodings in graph data is to use the Laplacian matrix of the graph. It is argued that the eigenvectors of the Laplacian matrix generalise the sinusoidal encodings to graphs Dwivedi & Bresson (2020).

311

320

312 3.3.1 EIGENVECTORS OF LAPLACIAN313

Let $G^{(U)} = (V, E)$, be an undirected graph without self loops, where V are the nodes and E the edges. The adjacency matrix $\mathbf{A}^{(U)} \in \mathbb{R}^{N \times N}$ is defined as $\mathbf{A}_{ij}^{(U)} = 1$ if $(i, j) \in E$ and $\mathbf{A}_{ij}^{(U)} = 0$ otherwise, with N = |V|. Given the degree of each node $d_i = \sum_{j=1}^{N} \mathbf{A}_{ij}^{(U)}$, the degree matrix is defined as $\mathbf{D} = \text{diag}(d_1, \dots, d_N)$. The (combinatorial) Laplacian and symmetrized Laplacian are defined as:

$$\mathbf{L}_{u} = \mathbf{D} - \mathbf{A}^{(U)} \in \mathbb{R}^{N \times N} \quad \mathbf{L}_{s} = \mathbf{D}^{-1/2} \mathbf{L} \mathbf{D}^{-1/2} \in \mathbb{R}^{N \times N}$$
(5)

Since, the \mathbf{L}_u of an undirected graph is real symmetric, \mathbf{L}_u can be decomposed as $\mathbf{L}_u = \mathbf{U}\mathbf{\Lambda}\mathbf{U}^T$, where U is the orthonormal matrix of eigenvectors $\mathbf{U} = v_0, \dots, v_{N-1}$ and $\mathbf{\Lambda}$ is the diagonal matrix of eigenvalues $\mathbf{\Lambda} = \lambda_0, \dots, \lambda_{N-1}$. Given the ordered eigenvalues $\lambda_0 \leq \dots \leq \lambda_{N-1}$, and the corresponding eigenvectors v_0, \dots, v_{N-1} , it always holds that $\lambda_0 = 0$ and $v_0 = \mathbf{1}$. In Beaini et al. (2021), it was shown that using only the eigenvector v_1 is enough to distinguish graphs not distinguishable by the 1-WL test. Additionally, in Kreuzer et al. (2021), the authors suggest that transformers with the full set of eigenvectors are universal approximators of the graph isomorphism problem.

328 **Example 3.1.** Let us consider the constraint-based model of Figure 3. The graph in Figure 4a shows the equivalent unipartite graph where the direction of the network has been omitted. In this graph, 330 the intensity of the color of the nodes is proportional to the value of the second eigenvector of the 331 combinatorial Laplacian v_1 . Generally, it is known that the second eigenvector is able to represent 332 closeness in the network, this is, nodes that are close to each other in the network will have similar values and distant nodes will have a larger difference in their values. In this sense, the combinatorial 333 Laplacian is able to provide information of the global positioning of a node in a network. Recall 334 that directionality is missed in the combinatorial Laplacian. However, the second eigenvector is 335 able to represent some redundancies of the network. For instance, in this graph, it happens that 336 nodes r_6 and r_7 , which belong to a redundant path, have similar values in the second eigenvector (i.e. 337 $v_{1,r6} \approx v_{1,r_7}$). The same case happens with nodes r_1 and r_2 (i.e. $v_{1,r1} \approx v_{1,r2}$). 338

339 3.3.2 MAGNETIC LAPLACIAN 340

348

As it was pointed out in Furutani et al. (2020) and Geisler et al. (2023), the combinatorial Laplacian fails to distinguish directionality on graphs. To address this issue, the authors propose the use of the eigenvectors of the magnetic Laplacian, where in addition to the connectivity of the network, the directionality of the network is encoded in the complex plane. Let $G^{(D)} = (V, E)$ be a directed graph with adjacency matrix $\mathbf{A}^{(D)}$, where $\mathbf{A}_{ij}^{(D)} = 1$ if $(i, j) \in E$ and $\mathbf{A}_{ij}^{(D)} = 0$ otherwise. Additionally, let $\mathbf{A}^{(U)}$ be the adjacency matrix of the undirected version of $G^{(D)}$. The magnetic Laplacian is defined as:

$$\mathbf{L}_m = \mathbf{D} - \mathbf{\Gamma}_{\mathbf{m}} \odot \mathbf{A}^{(U)} \in \mathbb{C}^{N \times N}$$
(6)

where \odot is the Hadamard product and $\Gamma_{\mathbf{m}} \in \mathbb{C}^{N \times N}$ is a Hermitian matrix whose (i, j) is equal to: $e^{i2\pi q(\mathbf{A}_{ij}^{(D)} - \mathbf{A}_{ji}^{(D)})}$, where $q \in [0, 1)$ is a rotation parameter or potential. Notice that when q = 0, the magnetic Laplacian is equal to the combinatorial Laplacian. In addition, notice that $e^{i2\pi q(\mathbf{A}_{ij}^{(D)} - \mathbf{A}_{ji}^{(D)})}$ encodes the direction of the edge (i, j). For undirected edges, i.e. $\mathbf{A}_{ij}^{(D)} = \mathbf{A}_{ji}^{(D)} = 1$, the value is equal to 1. For directed edges, i.e. $\mathbf{A}_{ij}^{(D)} = 1$, $\mathbf{A}_{ji}^{(D)} = 0$ and $\mathbf{A}_{ij}^{(D)} = 0$, $\mathbf{A}_{ji}^{(D)} = 1$, the value is equal to $e^{i2\pi q}$ and $e^{-i2\pi q}$ respectively. From here onwards, we will assume the eigenvalues of the magnetic Laplacian to be ordered: $0 = \lambda_0 \le \lambda_1 \le \cdots \le \lambda_{N-1}$.

Example 3.2. Let us consider again the constraint-based model of Figure 3. The graph in Figure 4b shows the equivalent unipartite graph. In this graph, the size of the nodes is proportional to the real part of the first eigenvector of the magnetic Laplacian, denoted $\operatorname{Re}(\phi_0^{(q)})$, and the color of the nodes is proportional to the imaginary part $\operatorname{Im}(\phi_0^{(q)})$. As it is shown in Furutani et al. (2020); Geisler et al. (2023), and as it can be seen in the graph, the magnetic Laplacian provides a topological sorting of the network in the complex plane. This is particularly useful in the case of metabolic networks, as it is able to capture redundancies in the network while also considering the directionality of the network.

For instance, notice that, in this graph, nodes r_1 and r_2 , which are both part of a redundant path, have the same color and size (i.e. $\operatorname{Re}(\phi_{0,r_1}^{(q)}) \approx \operatorname{Re}(\phi_{0,r_2}^{(q)})$ and $\operatorname{Im}(\phi_{0,r_1}^{(q)}) \approx \operatorname{Im}(\phi_{0,r_2}^{(q)})$). The same case happens with nodes r_3 , r_4 and r_5 . Notice also that, unlike the combinatorial Laplacian, the magnetic Laplacian is able to capture the fact that paths $m_3 \to r_g$ and $m_3 \to r_8$ are parallel.

370 3.3.3 NOTATION

To denote the set of eigenvalues and eigenvectors of the magnetic Laplacian, we will use the following notation: $\Lambda^{(q)} = \lambda_0^{(q)}, \ldots, \lambda_{N-1}^{(q)}$ and $\Phi^{(q)} = \phi_0^{(q)}, \ldots, \phi_{N-1}^{(q)}$, where $\phi_i^{(q)}$ is the *i*-th eigenvector of the magnetic Laplacian for rotation parameter *q*. We use $\phi_{i,j}^{(q)}$ to denote the entry of the *i*-th eigenvector corresponding with the *j*-th node.

We will use the fact that the magnetic Laplacian with q = 0 is equivalent to the combinatorial Laplacian to denote the eigenvalues and eigenvectors of the combinatorial Laplacian as $\Lambda^{(0)} =$ $\lambda_0^{(0)}, \ldots, \lambda_{N-1}^{(0)}$ and $\Phi^{(0)} = \phi_0^{(0)}, \ldots, \phi_{N-1}^{(0)}$ respectively. Similarly, we denote $\phi_{i,j}^{(0)}$ as the entry of the *i*-th eigenvector of the combinatorial Laplacian corresponding with the *j*-th node.

4 ARCHITECTURE

378379380381382

In this section, we present the organization of the proposed metabolic transformer. The architecture is a directed graph transformer, which is composed
of a positional encoding layer (that may include directional information)
and a graph transformer layer with directed message passing.

392 This architecture is motivated by the 393 fact that message passing alone is not 394 enough to identify the sufficient condi-395 tions that were introduced in Section 396 3. In Appendix B.1 an example is provided where the asymptotic com-397 plexity of identifying redundancies in 398 a network is of O(D) message pass-399 ing steps, with D being the diameter 400 of the network. Another significant 401 property of this architecture is that it 402 is able to capture directionality in the 403 network. Again, it is easy to see that 404



(a) Directed graph transformer (b) Directed graph transformer based on the SAT architecture based on the GPS architecture Chen et al. (2022). Rampášek et al. (2022).

Figure 5: Two proposed architectures for the directed graph transformer.

identifying redundancies in a network requires directional information. An example of this is provided
 in Appendix B.2.

4.1 SPECTRAL ENCODING

As mentioned before, bipartite metabolic networks will be treated as non-bipartite networks. This will allow us to compute spectral positional encodings as described before. This procedure is detailed in Appendix G. In Section 3 we described how the eigenvalues and eigenvectors of the combinatorial Laplacian and the magnetic Laplacian capture information about the global position of nodes in a graph. To be able to learn useful representations from this information, we will use the Laplacian Positional Encoder (LapPE), introduced in Kreuzer et al. (2021) and the Magnetic Laplacian Positional Encoder (MagLapPE), introduced in Geisler et al. (2023). A detailed description of these positional encoders is provided in Appendix C.

417 418

419

406 407

408

4.2 METABOLIC TRANSFORMER

Now that the positional encoding layers are introduced, we are ready to introduce the metabolic transformer layer. The approach taken is a combination of a directed message passing layer to capture the local structure of the network, and a self-attention layer over the whole set of nodes to capture global relations in the network. This combination of mesage passing with global attention has been presented before Chen et al. (2022); Rampášek et al. (2022), with promising results in many graph-related tasks. These works however, were limited to undirected graphs. In this work, two different flavours of graph transformers have been extended to include directed message passing.

427 428

429

430

431

1. The first approach (SAT), follows the work of Chen et al. (2022) and can be seen in Figure 5a. This approach is intended to capture structural similarities through the attention mechanism. The layer achieves this by using the GNN output as the query and the key of the attention mechanism, and the graph embeddings as the value. The output of the attention mechanism is used then as the input of the next layer. Formally, the l-th layer of the graph transformer

is defined as follows:

$$\mathbf{X}_{G}^{(l+1)} = \text{Dir-GNN}(\mathbf{X}^{(l)}, \mathbf{E}^{(l)}, \mathbf{A}^{(D)})$$
$$\mathbf{X}_{A}^{(l+1)} = \text{MHA}(\mathbf{X}_{G}^{(l+1)}, \mathbf{X}_{G}^{(l+1)}, X^{(l+1)})$$
$$\mathbf{X}^{(l+1)} = \text{Norm}(\text{FFN}(\mathbf{X}_{A}^{(l+1)} + \mathbf{X}^{(l)}))$$
(7)

where $\mathbf{X}^{(l+1)} \in \mathbb{R}^{N \times d}$ is the output of the *l*-th layer, $\mathbf{A}^{(D)}$ is the adjacency matrix of the graph, $\mathbf{E}^{(0)}$ are the initial edge features, and $\mathbf{X}^{(0)}$ are the sum of the initial graph features and positional encoder features.

2. The second approach (GPS), presented in Rampášek et al. (2022), is depicted in Figure 5b. This approach applies simultaneously a GNN layer, that captures the local information of the network, and a self-attention layer over the whole set of nodes embeddings. The output of both layers is added and the output of the layer is passed as input to the next layer. Formally, the l-th layer of the graph transformer is defined as follows:

$$\begin{aligned} \mathbf{X}_{G}^{(l+1)} &= \text{Dir-GNN}(\mathbf{X}^{(l)}, \mathbf{E}^{(l)}, \mathbf{A}^{(D)}) \\ \mathbf{X}_{A}^{(l+1)} &= \text{MHA}(\mathbf{X}^{(l)}) \\ \mathbf{X}^{(l+1)} &= \text{Norm}(\text{FFN}(\mathbf{X}_{G}^{(l+1)} + \mathbf{X}_{A}^{(l+1)} + \mathbf{X}^{(l)})) \end{aligned}$$
(8)

where $\mathbf{X}^{(l+1)} \in \mathbb{R}^{N \times d}$ is the output of the *l*-th layer, $\mathbf{A}^{(D)}$ is the adjacency matrix of the graph, $\mathbf{E}^{(0)}$ are the initial edge features, and $\mathbf{X}^{(0)}$ are the sum of the initial graph features and positional encoder features.

5 Results

This section reports the results ob-tained in a dataset of 23k genome-scale models with an average of 2k node. Since no previous dataset ex-ists for GEM, we are releasing the cu-rated dataset along with this publica-tion. The detailes of the dataset cura-tion can be found in Section E. The re-sults presented here include those obtained with the proposed graph trans-former models with the different po-sitional encoders. We are particularly interested in studying the effect of the different positional encoders and dif-ferent GNN layers on the overall per-formance. To do so, we compare the results obtained with baseline GNN models, which include: Graph Con-volutional Networks (GCN) Kipf &

Model	Params	Epoch time (s)	F1 ↑
SAT+LapPE+Dir-GCN	298393	619.71	0.6622
SAT+LapPE+Dir-GAT	349849	809.96	0.6328
SAT+LapPE+Dir-GINE	398209	757.61	0.5651
SAT+LapPE+Dir-Gated	498049	824.91	0.6628
SAT+MagLapPE+Dir-GCN	269145	614.65	0.6370
SAT+MagLapPE+Dir-GAT	320601	813.43	0.6786
SAT+MagLapPE+Dir-GINE	368961	735.50	0.5877
SAT+MagLapPE+Dir-Gated	468801	639.33	0.6974
GPS+LapPE+Dir-GCN	298393	557.39	0.6659
GPS+LapPE+Dir-GAT	349849	646.62	0.6718
GPS+LapPE+Dir-GINE	398209	587.89	0.6195
GPS+LapPE+Dir-Gated	498049	940.53	0.7145
GPS+MagLapPE+Dir-GCN	269145	546.68	0.6589
GPS+MagLapPE+Dir-GAT	320601	712.57	0.6708
GPS+MagLapPE+Dir-GINE	368961	735.50	0.5984
GPS+MagLapPE+Dir-Gated	468801	657.18	0.6841

Table 1: Results of the different models on the test set.

Welling (2016), Graph Attention Networks (GAT) Veličković et al. (2017), Graph Isomorphism
Networks (GINE) Xu et al. (2018a), and Gated Graph Neural Networks (Gated) Li et al. (2015). For
each of the baseline models, we used the directed version of the model, as described in Section 4.
Details on the hyperparameters and experimental setup can be found in Appendix F.

Model	F1 (+) ↑	ROC AUC (+) \uparrow	F1 (-) ↑	ROC AUC (-) ↑
FlowGAT Hasibi et al. (2024)	0.85 ± 0.033	0.495 ± 0.053	0.02 ± 0.04	0.525 ± 0.075
Metabolic Transformer (ours)	0.845 ± 0.027	0.857 ± 0.022	0.69 ± 0.026	0.849 ± 0.017

Table 2: Performance comparison in synthetic reaction essentiality prediction on *E.coli* model
iML1515 Monk et al. (2017). Here (+) indicates essentiality as positive class, while (-) indicates the
opposite problem, this is, prediction of non-essentiality.

486	Model	F1 (+) ↑	ROC AUC (+) ↑	F1 (-) ↑	ROC AUC (-) ↑
487	FlowGAT Hasibi et al. (2024)	0.86 ± 0.015	0.56 ± 0.121	0.161 ± 0.099	0.587 ± 0.134
488	Metabolic Transformer (ours)	0.68 ± 0.031	0.845 ± 0.03	0.624 ± 0.093	0.797 ± 0.083



As it is explained in Appendix E, the generated dataset is highly imbalanced, with an average of 495 8.8% of positive samples. To account for this, we report the F1 score of the models to compare 496 their performance. From the results shown in Table 1, we can see that there is a clear dependence of 497 the results on the message passing scheme used. Overall, the results suggest that message passing 498 layers that aggregate the messages in a non-linear way (Dir-GAT, Dir-Gated) perform better than 499 those that weight each message contribution the same (Dir-GCN, Dir-GINE). This suggests that the 500 solution suffers from the well-known over-squashing problem Di Giovanni et al. (2023). Regarding 501 the graph transformer architecture, generally, GPS-based solutions seem to perform slightly better 502 than SAT-based solutions. If we compare now the results obtained with the different positional encoders, surprisingly, there is no clear advantage of using one over the other. It seems that the MagLapPE performs better with the SAT-based solutions, while the LapPE performs better with the 504 GPS-based solutions. The best results are obtained with the GPS + LapPE + Dir-Gated model. Recall 505 that LapPE does not capture the directional information of the graph. Despite this, LapPE is equally 506 competitive than MagLapPE and outperforms MagLapPE in the case mentioned. This might suggest 507 that (i) direction information is not as important for the task at hand or (ii) the information captured 508 with the message passing is enough to capture the directional information of the graph. These results 509 also suggest that, for this task, the use of the message passing scheme seems to be more important 510 than the positional encoding scheme.

In Appendix D we provide an ablation study
of the different components of the proposed architecture. A performance comparison between
GPU and CPU inference for essential reactions
computation is presented in Appendix H

493 494

In order to evaluate the performance of the 517 metabolic transformer against the existing state-518 of-the-art, we compared the results obtained 519 with the FlowGAT model Hasibi et al. (2024). 520 As metabolic transformer, here we used the best-521 performing model according to Table 1, this is, 522 GPS+LapPE+Dir-Gated. Table 2 show the re-523 sults obtained in synthetic knock-out essential 524 reaction prediction, using the iML1515 model 525 of Escherichia coli Monk et al. (2017). As men-526 tioned previously in this work, since FlowGAT relies in locality information, the capability to 527 model essential reactions is limited, which ex-528 plains the difference in performance. Table 3 529 shows the results obtained in real gene essen-530 tiality prediction using the same model. A fine-531 tuning was performed with the metabolic trans-532 former to adapt the model to the task at hand. The results show that FlowGAT shows a higher 534 F1 score, while performing poorly in the ROC AUC metric. In order to clarify this divergence 536 in scores, confusion matrices from both problems are shown in Figures 7 and 6. In this Figure it can be seen that FlowGAT tends to classify all 538 reactions as positive, while almost no discrimination is done with negative classes. Since the



(a) FlowGAT Hasibi et al. (b) Metabolic Transformer (2024) (ours)

Figure 6: Confusion matrix for synthetic reaction essentiality prediction on *E.coli* model iML1515 Monk et al. (2017), obtained from best F1-score.



(a) FlowGAT Hasibi et al. (b) Metabolic Transformer (2024) (ours)

Figure 7: Confusion matrix for real gene essentiality prediction on *E.coli* model iML1515 Monk et al. (2017), obtained from best F1-score.

postive class here is the majority class, the F1 score obtained is large, while the ROC AUC shows that
discriminative power of FlowGAT is considerably lower than the metabolic transformer introduced in
this work. We note that the evaluation dataset used consists of a small number of samples with a high
presence of positive samples, which explains the fluctuation in the ROC AUC metric. Note also that,
while FlowGAT is a specialized model narrowed down for this particular tasks, our approach is a
general model finetuned for gene essentiality prediction.

546 547

548

6 CONCLUSIONS

549 This work proposes the metabolic trasnformers, a novel approach for the prediction of essential 550 reactions in genome-scale models by using directed graph transformers, entirely trained with synthetic 551 knock-out data. The proposed approach is based on the use of directed GNNs as the base model to 552 capture local information, and the use of positional encodings based on the combinatorial Laplacian 553 and the magnetic Laplacian. To compare the performance of the proposed approach, we built a dataset of genome-scale models from the metabolism of different microorganisms from public databases. 554 Aditionally, we compared the performance of the proposed approach with existing methods for 555 essential reactions and genes prediction, and we showed that the proposed approach achieves state-of-556 the-art performance in both tasks. The study on the architectural choices show that the performance 557 is highly dependent on the message passing scheme used. In particular, the best results are obtained 558 with the use of gated GNNs with a GPS architecture and Laplacian-based positional encodings. The 559 results also show no clear advantage of using magnetic Laplacian-based positional encodings over 560 Laplacian-based positional encodings, when combined with directed message passing. This work, 561 however, is fundamentally limited by the scalability of the model, both from a data and memory 562 perspective. Transformers are well known to have a $O(N^2)$ memory complexity. In our case, we 563 used graphs from 113 microorganisms with an average of 2190 nodes, with the largest one having 564 5861 nodes. Currently, the largest available genome-scale model is the human model Recon3D Brunk et al. (2018) with 17683 nodes, which poses a challenge to the scalability of the model. Possible 565 solutions to this issue include using more efficient attention mechanisms Choromanski et al. (2020) 566

7 ETHICS STATEMENT

568 569

567

The contribution of this work is to provide a novel approach to identify essential reactions in GEMs. This has the potential to facilitate the identification of essential reactions in a broad range of automatically reconstructed GEMs, a task previously deemed infeasible or requiring extensive manual curation of models. Current approaches that identify essential reactions in GEMs are aimed at contributing to the development of new antibiotics or cancer treatments, among others. Therefore

we do not expect any ethical or societal issues to arise from this work.

575 576 577

578 579

580

581

582 583

584

8 REPRODUCIBILITY STATEMENT

Both the code and datasets required for the reproducibility of the results reported will be made available under GNU-GPLv3 license upon acceptance of the paper. Additionally, the steps taken to generate the dataset and hyperparameters used can be found in the Appendix F, E and G.

References

- Towards a widespread adoption of metabolic modeling tools in biopharmaceutical industry: a process systems biology engineering perspective. *npj Systems Biology and Applications*, 6(1):1–5, dec 2020. ISSN 20567189. doi: 10.1038/s41540-020-0127-y. URL https://doi.org/10.1038/s41540-020-0127-y.
- ⁵⁸⁹ Uri Alon and Eran Yahav. On the bottleneck of graph neural networks and its practical implications.
 arXiv preprint arXiv:2006.05205, 2020.
- Dominique Beaini, Saro Passaro, Vincent Létourneau, Will Hamilton, Gabriele Corso, and Pietro
 Liò. Directional graph networks. In *International Conference on Machine Learning*, pp. 748–758.
 PMLR, 2021.

623

628

629

630 631

632

633

634 635

636

637

641

594	Devu Bo, Chuan Shi, Lele Wang, and Renije Liao. Specformer: Spectral graph neural networks meet
595	transformers. arXiv preprint arXiv:2303.01028, 2023.
596	

- Elizabeth Brunk, Swagatika Sahoo, Daniel C Zielinski, Ali Altunkaya, Andreas Dräger, Nathan 597 Mih, Francesco Gatto, Avlant Nilsson, German Andres Preciat Gonzalez, Maike Kathrin Aurich, 598 et al. Recon3d enables a three-dimensional view of gene variation in human metabolism. Nature biotechnology, 36(3):272–281, 2018. 600
- 601 Finja Büchel, Nicolas Rodriguez, Neil Swainston, Clemens Wrzodek, Tobias Czauderna, Roland 602 Keller, Florian Mittag, Michael Schubert, Mihai Glont, Martin Golebiewski, et al. Path2models: 603 large-scale generation of computational models from biochemical pathway maps. BMC systems biology, 7:1-19, 2013. 604
- 605 Binghong Chen, Chengtao Li, Hanjun Dai, and Le Song. Retro*: learning retrosynthetic planning 606 with neural guided a* search. In International conference on machine learning, pp. 1608–1616. 607 PMLR, 2020. 608
- Dexiong Chen, Leslie O'Bray, and Karsten Borgwardt. Structure-aware transformer for graph 609 representation learning. In International Conference on Machine Learning, pp. 3469–3489. PMLR, 610 2022. 611
- 612 Krzysztof Choromanski, Valerii Likhosherstov, David Dohan, Xingyou Song, Andreea Gane, Tamas 613 Sarlos, Peter Hawkins, Jared Davis, Afroz Mohiuddin, Lukasz Kaiser, et al. Rethinking attention 614 with performers. arXiv preprint arXiv:2009.14794, 2020. 615
- Francesco Di Giovanni, Lorenzo Giusti, Federico Barbero, Giulia Luise, Pietro Lio, and Michael M 616 Bronstein. On over-squashing in message passing neural networks: The impact of width, depth, 617 and topology. In International Conference on Machine Learning, pp. 7865–7885. PMLR, 2023. 618
- 619 Alexey Dosovitskiy, Lucas Beyer, Alexander Kolesnikov, Dirk Weissenborn, Xiaohua Zhai, Thomas 620 Unterthiner, Mostafa Dehghani, Matthias Minderer, Georg Heigold, Sylvain Gelly, et al. An 621 image is worth 16x16 words: Transformers for image recognition at scale. arXiv preprint 622 arXiv:2010.11929, 2020.
- Vijay Prakash Dwivedi and Xavier Bresson. A generalization of transformer networks to graphs. 624 arXiv preprint arXiv:2012.09699, 2020. 625
- 626 Ali Ebrahim, Joshua A Lerman, Bernhard O Palsson, and Daniel R Hyduke. Cobrapy: constraints-627 based reconstruction and analysis for python. BMC systems biology, 7:1–6, 2013.
 - Lilli J Freischem, Mauricio Barahona, and Diego A Oyarzún. Prediction of gene essentiality using machine learning and genome-scale metabolic models. bioRxiv, pp. 2022-03, 2022.
 - Satoshi Furutani, Toshiki Shibahara, Mitsuaki Akiyama, Kunio Hato, and Masaki Aida. Graph signal processing for directed graphs based on the hermitian laplacian. In *Machine Learning* and Knowledge Discovery in Databases: European Conference, ECML PKDD 2019, Würzburg, Germany, September 16–20, 2019, Proceedings, Part I, pp. 447–463. Springer, 2020.
 - Simon Geisler, Yujia Li, Daniel Mankowitz, Ali Taylan Cemgil, Stephan Günnemann, and Cosmin Paduraru. Transformers meet directed graphs. arXiv preprint arXiv:2302.00049, 2023.
- 638 Changdai Gu, Gi Bae Kim, Won Jun Kim, Hyun Uk Kim, and Sang Yup Lee. Current 639 status and applications of genome-scale metabolic models. Genome Biology, 20(1): 1-18, jun 2019. ISSN 1474760X. doi: 10.1186/s13059-019-1730-3. URL https: 640 //link.springer.com/articles/10.1186/s13059-019-1730-3https: //link.springer.com/article/10.1186/s13059-019-1730-3. 642
- 643 Will Hamilton, Zhitao Ying, and Jure Leskovec. Inductive representation learning on large graphs. 644 Advances in neural information processing systems, 30, 2017. 645
- Ramin Hasibi, Tom Michoel, and Diego A Oyarzún. Integration of graph neural networks and genome-646 scale metabolic models for predicting gene essentiality. *npj Systems Biology and Applications*, 10 647 (1):24, 2024.

648 649 650	Monika Heiner, David Gilbert, and Robin Donaldson. Petri nets for systems and synthetic biology. volume 5016, pp. 215–264, 06 2008. doi: 10.1007/978-3-540-68894-5_7.
651 652 653	Christopher S Henry, Matthew D Jankowski, Linda J Broadbelt, and Vassily Hatzimanikatis. Genome- scale thermodynamic analysis of escherichia coli metabolism. <i>Biophysical journal</i> , 90(4):1453– 1461, 2006.
654 655 656	Weihua Hu, Matthias Fey, Hongyu Ren, Maho Nakata, Yuxiao Dong, and Jure Leskovec. Ogb-lsc: A large-scale challenge for machine learning on graphs. <i>arXiv preprint arXiv:2103.09430</i> , 2021.
657 658 659	Michael T Judge, Yue Wu, Fariba Tayyari, Ayuna Hattori, John Glushka, Takahiro Ito, Jonathan Arnold, and Arthur S Edison. Continuous in vivo metabolism by nmr. <i>Frontiers in Molecular Biosciences</i> , 6:26, 2019.
660 661 662	Jinwoo Kim, Dat Nguyen, Seonwoo Min, Sungjun Cho, Moontae Lee, Honglak Lee, and Seunghoon Hong. Pure transformers are powerful graph learners. <i>Advances in Neural Information Processing Systems</i> , 35:14582–14595, 2022.
664 665 666 667	Zachary A King, Justin Lu, Andreas Dräger, Philip Miller, Stephen Federowicz, Joshua A Lerman, Ali Ebrahim, Bernhard O Palsson, and Nathan E Lewis. Bigg models: A platform for integrating, standardizing and sharing genome-scale models. <i>Nucleic acids research</i> , 44(D1):D515–D522, 2016.
668 669 670	Thomas N Kipf and Max Welling. Semi-supervised classification with graph convolutional networks. <i>arXiv preprint arXiv:1609.02907</i> , 2016.
671 672	Wouter Kool, Herke Van Hoof, and Max Welling. Attention, learn to solve routing problems! <i>arXiv</i> preprint arXiv:1803.08475, 2018.
673 674 675 676	Devin Kreuzer, Dominique Beaini, Will Hamilton, Vincent Létourneau, and Prudencio Tossou. Rethinking graph transformers with spectral attention. <i>Advances in Neural Information Processing</i> <i>Systems</i> , 34:21618–21629, 2021.
677 678 679	Qimai Li, Zhichao Han, and Xiao-Ming Wu. Deeper insights into graph convolutional networks for semi-supervised learning. In <i>Proceedings of the AAAI conference on artificial intelligence</i> , volume 32, 2018.
680 681 682	Yujia Li, Daniel Tarlow, Marc Brockschmidt, and Richard Zemel. Gated graph sequence neural networks. <i>arXiv preprint arXiv:1511.05493</i> , 2015.
683 684 685 686	Derek Lim, Joshua Robinson, Lingxiao Zhao, Tess Smidt, Suvrit Sra, Haggai Maron, and Stefanie Jegelka. Sign and basis invariant networks for spectral graph representation learning. <i>arXiv</i> preprint arXiv:2202.13013, 2022.
687 688	Ilya Loshchilov and Frank Hutter. Decoupled weight decay regularization. <i>arXiv preprint arXiv:1711.05101</i> , 2017.
689 690 691 692 693	Rahuman S Malik-Sheriff, Mihai Glont, Tung VN Nguyen, Krishna Tiwari, Matthew G Roberts, Ashley Xavier, Manh T Vu, Jinghao Men, Matthieu Maire, Sarubini Kananathan, et al. Biomodels—15 years of sharing computational models in life science. <i>Nucleic acids research</i> , 48(D1): D407–D415, 2020.
694 695 696	Erxue Min, Runfa Chen, Yatao Bian, Tingyang Xu, Kangfei Zhao, Wenbing Huang, Peilin Zhao, Junzhou Huang, Sophia Ananiadou, and Yu Rong. Transformer for graphs: An overview from architecture perspective. <i>arXiv preprint arXiv:2202.08455</i> , 2022.
697 698 699 700	Jonathan M Monk, Colton J Lloyd, Elizabeth Brunk, Nathan Mih, Anand Sastry, Zachary King, Rikiya Takeuchi, Wataru Nomura, Zhen Zhang, Hirotada Mori, et al. i ml1515, a knowledgebase that computes escherichia coli traits. <i>Nature biotechnology</i> , 35(10):904–908, 2017.
700	T. Murata. Petri Nets: Properties, Analysis and Applications. <i>Procs. of the IEEE</i> , 77(4):541–580, 1989. doi: 10.1109/5.24143.

702 703 704 705	Edward J O'brien, Joshua A Lerman, Roger L Chang, Daniel R Hyduke, and Bernhard Ø Palsson. Genome-scale models of metabolism and gene expression extend and refine growth phenotype prediction. <i>Molecular systems biology</i> , 9(1):693, 2013.
706 707 708	Jeffrey D Orth, Ines Thiele, and Bernhard Ø Palsson. What is flux balance analysis? <i>Nature Biotechnology 2010 28:3</i> , 28(3):245–248, mar 2010. ISSN 1546-1696. doi: 10.1038/nbt.1614. URL https://www.nature.com/articles/nbt.1614.
709 710 711	Jeffrey D Orth, Tom M Conrad, Jessica Na, Joshua A Lerman, Hojung Nam, Adam M Feist, and Bernhard Ø Palsson. A comprehensive genome-scale reconstruction of Escherichia coli metabolism– 2011. <i>Molecular Systems Biology</i> , 7(1), 2011. ISSN 1744-4292. doi: 10.1038/msb.2011.65.
712 713 714 715 716	Jelili Oyelade, Itunuoluwa Isewon, Efosa Uwoghiren, Olufemi Aromolaran, and Olufunke Oladipupo. In Silico Knockout Screening of Plasmodium falciparum Reactions and Prediction of Novel Essential Reactions by Analysing the Metabolic Network. <i>BioMed Research International</i> , 2018, 2018. doi: 10.1155/2018/8985718.
717 718 719	Ladislav Rampášek, Michael Galkin, Vijay Prakash Dwivedi, Anh Tuan Luu, Guy Wolf, and Do- minique Beaini. Recipe for a general, powerful, scalable graph transformer. <i>Advances in Neural</i> <i>Information Processing Systems</i> , 35:14501–14515, 2022.
720 721 722 723	Yu Rong, Yatao Bian, Tingyang Xu, Weiyang Xie, Ying Wei, Wenbing Huang, and Junzhou Huang. Self-supervised graph transformer on large-scale molecular data. <i>Advances in Neural Information</i> <i>Processing Systems</i> , 33:12559–12571, 2020.
724 725 726	Emanuele Rossi, Bertrand Charpentier, Francesco Di Giovanni, Fabrizio Frasca, Stephan Günnemann, and Michael Bronstein. Edge directionality improves learning on heterophilic graphs. <i>arXiv</i> preprint arXiv:2305.10498, 2023.
727 728 729	Pierre Salvy and Vassily Hatzimanikatis. The etfl formulation allows multi-omics integration in thermodynamics-compliant metabolism and expression models. <i>Nature communications</i> , 11(1):30, 2020.
730 731 732	Franco Scarselli, Marco Gori, Ah Chung Tsoi, Markus Hagenbuchner, and Gabriele Monfardini. The graph neural network model. <i>IEEE transactions on neural networks</i> , 20(1):61–80, 2008.
733 734 735 726	Amit Varma and Bernhard Ø. Palsson. Metabolic Flux Balancing: Basic Concepts, Scientific and Practical Use. <i>Nature Biotechnology</i> , 12(10):994–998, October 1994. ISSN 1087-0156. doi: 10.1038/nbt1094-994.
737 738 739	Ashish Vaswani, Noam Shazeer, Niki Parmar, Jakob Uszkoreit, Llion Jones, Aidan N Gomez, Łukasz Kaiser, and Illia Polosukhin. Attention is all you need. <i>Advances in neural information processing systems</i> , 30, 2017.
740 741 742	Petar Veličković, Guillem Cucurull, Arantxa Casanova, Adriana Romero, Pietro Lio, and Yoshua Bengio. Graph attention networks. <i>arXiv preprint arXiv:1710.10903</i> , 2017.
743 744 745	Mingjian Wen, Evan Walter Clark Spotte-Smith, Samuel M Blau, Matthew J McDermott, Aditi S Krishnapriyan, and Kristin A Persson. Chemical reaction networks and opportunities for machine learning. <i>Nature Computational Science</i> , 3(1):12–24, 2023.
746 747 748	Keyulu Xu, Weihua Hu, Jure Leskovec, and Stefanie Jegelka. How powerful are graph neural networks? <i>arXiv preprint arXiv:1810.00826</i> , 2018a.
749 750 751	Keyulu Xu, Chengtao Li, Yonglong Tian, Tomohiro Sonobe, Ken-ichi Kawarabayashi, and Stefanie Jegelka. Representation learning on graphs with jumping knowledge networks. In <i>International conference on machine learning</i> , pp. 5453–5462. PMLR, 2018b.
752 753 754	Zhihui Yang, Juan Liu, Hayat Ali Shah, and Jing Feng. A novel hybrid framework for metabolic pathways prediction based on the graph attention network. <i>BMC bioinformatics</i> , 23(5):1–15, 2022.
755	Jiaxuan You, Zhitao Ying, and Jure Leskovec. Design space for graph neural networks. Advances in Neural Information Processing Systems, 33:17009–17021, 2020.

756	Guido Zampieri, Supreeta Vijavakumar, Elisabeth Yaneske, and Claudio Angione. Machine and deep
757	learning meet genome-scale metabolic modeling. <i>PLoS computational biology</i> , 15(7):e1007084,
758	2019.
759	
760	
761	
762	
763	
764	
765	
766	
707	
760	
709	
771	
779	
773	
774	
775	
776	
777	
778	
779	
780	
781	
782	
783	
784	
785	
786	
787	
788	
789	
790	
791	
792	
793	
794	
795	
797	
798	
799	
800	
801	
802	
803	
804	
805	
806	
807	
808	
809	



813 814 815

816

817 818 819

820

810

811 812

Figure 8: Example Petri net where r_1 and r_2 belong to redundant pahts. A total of 2N+2 message passing steps are required to reach r_2 from r_1 .

A TRANSFORMERS

Transformers are neural network models that have been shown to be very effective in a wide range of tasks, ranging from language modelling Vaswani et al. (2017), to computer vision Dosovitskiy et al. (2020) or approximating combinatorial problems such as the Travelling Salesman Problem Kool et al. (2018). Transformers are composed of two main blocks: self-attention and feed-forward layers. The self-attention layer is a function that takes as input the *N* node features $\mathbf{X} \in \mathbb{R}^{N \times d}$, it linearly projects \mathbf{X} into the query \mathbf{Q} , key \mathbf{K} and value \mathbf{V} this is, $\mathbf{Q} = \mathbf{X}\mathbf{W}_Q$, $\mathbf{K} = \mathbf{X}\mathbf{W}_K$ and $\mathbf{V} = \mathbf{X}\mathbf{W}_V$, and computes self-attention as:

SelfAttention(
$$\mathbf{Q}, \mathbf{K}, \mathbf{V}$$
) = softmax $\left(\frac{\mathbf{Q}\mathbf{K}^T}{\sqrt{d}}\right)\mathbf{V} \in \mathbb{R}^{N \times d}$ (9)

where d is the dimension of \mathbf{Q} and \mathbf{W}_Q , \mathbf{W}_K , $\mathbf{W}_V \in \mathbb{R}^{d \times d}$ are trainable weight matrices. Generally, multi-head attention is used, this is, the self-attention is computed h times with different weight matrices, and the resulting outputs are concatenated. The multi-head attention layer is then followed by a feed-forward layer. The overall transformer architecture can be defined as:

828

829 830

831

832

835 836 837

838

839 840

841

842

845 846

847

 $\mathbf{X}' = \mathbf{X} + MHA(\mathbf{X})$ Transformer(\mathbf{X}) = FFN(\mathbf{X}') = ReLU($\mathbf{X}'\mathbf{W}_1$) \mathbf{W}_2 (10)

where MHA is the multi-head attention layer, ReLU is the activation function, and $\mathbf{W}_1, \mathbf{W}_2 \in \mathbb{R}^{d \times d}$ are trainable weight matrices. In the above equation, MHA($\mathbf{X}, \mathbf{X}, \mathbf{X}$) has been written as MHA(\mathbf{X}) for simplicity.

B CHALLENGES

This section provides an insight into the existing challenges to identify redundancies in metabolic networks, and how these challenges have driven our architectural decisions.

B.1 LONG-RANGE DEPENDENT

It is straightforward to check that the message passing framework is constrained when it comes to 848 identifying redundancies on networks. To see this, consider the example network of Figure 8. In this 849 network, reaction r_s is a source reaction in the network, and reaction r_q is the objective reaction. It 850 is clear that there are two different paths from r_s to r_q , one path that goes through reaction r_1 and 851 another path that goes through reaction r_2 . This makes reactions r_1 and r_2 not essential. Notice that 852 between reaction r_1 and r_2 there are a total of 2N+2 nodes. This means that, if we wanted to transfer 853 information from r_1 to r_2 using message passing, we would need at least 2N + 2 steps (assuming 854 undirected message passing). Ultimately, this means that, for reaction r_1 to acknowledge the existence 855 of reaction r_2 , it would need to pass information through at least 2N + 2 steps. Therefore, it is clear that the number of steps required to identify redundancies in the network scales linearly with the 856 number of nodes in the network, this is, given a network N, the worst case number of steps required 857 is O(D), where D is the diameter of N. 858

Given this limitation of message passing, we need to look for a different mechanism for capturing
the global information of the network. This shortcoming in message passing may be addressed by
using positional encodings Dwivedi & Bresson (2020). Since positional encodings try to capture the
global position of a node in the network, they may reduce the number of steps required to identify
redundancies in the network. However, as shown in Section D, GNNs with positional encodings are
not powerful enough to identify redundancies in the network.



 r_2 is essential.

870

tions.

871 872

873

874

875 876

877

892

893 894

895

896

897

898 899

900

Figure 9: The directionality of the network determines the essentiality of the reactions. Both networks in Figures (a) and (b) share the same undirected topology, but the directionality of the edges is different. This makes reaction r_1 essential in Figure (a) and not essential in Figure (b).

tions r_1 and r_2 are essential reaction r_1 is not essential and reaction ogy matches the one of Figures (a)

and (b).

B.2 DIRECTIONALITY

Given the wide application of graph learning methods to undirected graphs, one might be tempted to use undirected graphs to represent metabolic networks. Again, it is straightforward to show that the directionality of the network is a key property in the identification of redundancies and that ignoring it leads to a loss of information in the network. To see this, consider the Petri nets of Figures 9a and 9b, where reaction r_g is the objective reaction. Notice that, given the direction of the reactions, reaction r_1 is essential in Figure 9a but not essential in Figure 9b, since in the last, alternative reactions exist that are able to supply the objective reaction. Notice that, if we ignore the directionality of the reactions, both networks share the same topology, depicted in Figure 9c.

In this graph, reaction r_2 is always involved in a path that leads to the objective reaction r_g . Therefore, it is easy to infer that this reaction will always be essential, even if we ignore the directionality of the reactions. However, without directionality, it is impossible to infer whether reaction r_1 is essential or not. It is clear then, that any approach that dismisses the direction of the network will suffer a loss of key information.

C SPECTRAL ENCODING

This section provides a detailed description of the positional encoding proposed in Section 4. In particular, we will definitions are provided for the combinatorial Laplacian Positional Encoder (LapPE) Kreuzer et al. (2021) and the Magnetic Laplacian Positional Encoder (MagLapPE) Geisler et al. (2023).

C.1 LAPLACIAN POSITIONAL ENCODER

To encode the spectral information of the combinatorial Laplacian we use an approach similar to the one in Kreuzer et al. (2021). In this work, the first k eigenvectors with the minimum eigenvalues are selected. These eigenvectors are concatenated with their corresponding eigenvalues, passed through a linear layer, a self-attention layer and finally each embedding is pooled node-wise through a sum pooling. More formally, given the k eigenvectors corresponding to the node i, which we denote as $\phi_{:k,i}$, and the k eigenvalues, which we denote as $\lambda_{:k}$, the Laplacian Positional encoding for node i is defined as follows:

 $X_{i} = \text{FFN}\left(\begin{bmatrix}\phi_{:k,i}\\\lambda_{:k}^{T}\end{bmatrix}\right)$ $X'_{i} = \text{MHA}(X_{i})$

 $X_i'' = \text{FFN}\left(\sum_{j}^k X_{i,j}'\right)$

(11)

- 908
- 909 910

10

911

912

913 914

915

where $X_i'' \in \mathbb{R}^d$ is the final embedding of the node *i*, with *d* being the dimension of the embedding. Normalization has been omitted from the above definition. To handle the sign invariance of the eigenvectors, the sign of $\phi_{:k,i}$ is randomly flipped, as it was done in Dwivedi & Bresson (2020).

918 C.2 MAGNETIC LAPLACIAN POSITIONAL ENCODER 919

920 To encode the spectral information of the magnetic Laplacian, we will use the positional encoder proposed in Geisler et al. (2023). Similarly as in the Laplacian Positional Encoder, the first k921 eigenvectors with the minimum eigenvalues are selected. Let us denote, $\phi_{k,i}$ as the k eigenvectors 922 corresponding to the node i, and $\lambda_{:k}$ as the k eigenvalues. The Magnetic Laplacian Positional Encoder 923 is defined as follows: 924

> $X_i = \text{FFN} \left(\text{Re}(\phi_{:k,i}) \parallel \text{Im}(\phi_{:k,i}) \right) \parallel \lambda_{:k}^T$ $X_i' = \mathsf{MHA}(X_i) + X_i$ $\mathbf{X}'' = \operatorname{FFN}\left(\prod_{i=1}^{N} X'_i \right)$ (12) $\mathbf{X}''' = \mathrm{MHA}(\mathbf{X}'') + \mathbf{X}''$

930 931 932

934 935

936

937

938

939

940

941 942

943 944

945

946

953

968

925

926

927

928 929

where $\mathbf{X}'' \in \mathbb{R}^{N \times d}$ is the final matrix of embeddings of the N nodes, with d being the dimension of 933 the embedding. Normalization and dropout have been omitted from the definition. Notice that, unlike the Laplacian Positional Encoder, this encoder includes skip connections after the self-attention layer, and performs self-attention before and after concatenating the embeddings of the nodes. Empirically, we did not observe any significant advantage of using self-attention. To handle the sign invariance of the eigenvectors, we use the same approach proposed in Geisler et al. (2023), this is, the sign of each eigenvector is determined such that the maximum real magnitude is positive. The authors also propose the use of SignNet Lim et al. (2022) to handle the sign invariance, however, empirically we did not observe any significant advantage using SignNet. The eigenvector normalization is performed as described in the original paper Geisler et al. (2023).

D ABLATION STUDY

In this section, we perform an ablation study to show the importance of the different components of the proposed model. In particular, we will show the performance obtained without the use of message passing, without the use of transformers, and the use of directed GNNs.

Model	Params	Epoch time (s)	F1 ↑
Transformer + MagLapPE	482305	548.12	0.4166

Table 4: Ablation without message passing.

954 First, let us consider the model without the use of message passing, this is, using only a transformer 955 layer and the positional encodings. Here we used directly magnetic Laplacian positional encodings, 956 since combinatorial Laplacian positional encodings are not able to capture directionality. The results are shown in Table 4. Here we can see that the model performance is not as good, with an F1 score of 957 0.4166. Based on this, it seems that the model struggles to capture local information relying only on 958 the positional encodings. 959

Model	Params	Epoch time (s)	F1 ↑
SAT + LapPE + Gated	374785	642.65	0.6003
GPS + LapPE + Gated	374785	581.06	0.6006
SAT + MagLapPE + Gated	345537	607.56	0.6022
GPS + MagLapPE + Gated	345537	553.92	0.5524

Table 5: Ablation using non-directed message passing.

Let us now consider the model without directed GNNs. Here we used both the SAT and the GPS 969 architecture with an undirected gated GNN layer. We resort to the gated GNN layer since it shows 970 a better performance than other GNN layers in our given dataset. The results are shown in Table 5. 971 Here we can see that the model shows a better performance, reaching up to 0.6022 F1 score. Notice

that this outperforms some of the models presented in Table 1. However, it is still far from the best performance obtained on the task. Compared to the results in Table 4, there is a clear improvement in the score. This suggests that a great performance can be obtained in the task even without the directionality information.

Model	Params	Epoch time (s)	F1 ↑
LapPE + Dir-Gated	662785	167.62	0.6445
MagLapPE + Dir-Gated	591169	166.88	0.6512

980 981

982

Table 6: Ablation without transformer.

983 Finally, let us consider the model without the use of transformers. Here we used directly a sequence 984 of GNN layers without any transformer layer. For this task, we used a directed gated GNN layer. The 985 results are shown in Table 6. Here we can see that the model shows a surprisingly good performance, 986 reaching up to 0.6512 F1 score. This performance again outperforms some of the models presented 987 in Table 1, however, it is still far from the best performance obtained on the task. This highlights the 988 importance of the message passing layer and, in particular, the importance of the directed GNN layer. 989 Finally, it shows that, without the global information provided by the transformer, the model is still 990 not able to fully capture the redundancies in the network.

991 992

993

E DATASET

994 To generate the dataset used in this work, we gathered 197 genome-scale models from the BiGG 995 King et al. (2016) and Biomodels Malik-Sheriff et al. (2020) databases. The models correspond 996 with a total of 113 different organisms, having multiple strains for some of them. The amount of metabolites of the models ranges from 7 to 2038 metabolites with an average of 772. The amount 997 of reactions ranges from 6 to 4047 reactions with an average of 1115. The largest model contains 998 a total of 5861 nodes. The number of essential reactions is on average 26% of the total reactions 999 with a standard deviation of 0.17. If we compare the number of essential reactions with the total 1000 number of nodes, on average, 14% of the nodes are essential reactions with a standard deviation of 1001 0.14. It can be seen that the amount of essential reactions is highly imbalanced, with an average of 1002 218 essential reactions. To generate our dataset the models were split into 153 models for training, 21 1003 models for validation and 21 models for testing. Since models of the same organism tend to share 1004 more similar network topologies, models of the same organism were not split into different sets. The 1005 split between training, validation and test sets was done manually to ensure that the models in all sets have a balanced number of reactions, metabolites, and essential reactions, as well as the maximum 1007 degree of the nodes and the degree of the objective reaction. The detailed list of models and splits is 1008 available as supplementary material.

1009

Essential reactions computation To compute the essential reactions of each model, we used the COBRApy Ebrahim et al. (2013) framework. For all the models, we remove restrictions that enforce a non-null flux in the reactions (i.e. we set L = 0), this is, we assume a rich medium and we impose no constraint on growth production or any other metabolite. Notice that, the proposed method in this work does not limit the possibility of adding such restrictions for reactions. Future approaches could include reaction fluxes as additional features.

1016 **Data augmentation** Since we are able to compute essential reactions of any model, we performed 1017 random modifications in the models as a way to augment the dataset. To generate a modified model, 1018 we performed randomly two types of modifications: (i) we randomly chose any reaction of the 1019 model and used it as the objective reaction and (ii) we took the biomass reaction and randomly 1020 added or removed between 1 and 10 reactants and between 1 and 10 products of the model. In 1021 the case of producing an infeasible problem, the generated model was discarded. Notice that both 1022 presented modifications alter the objective function and therefore the computed essential reactions. 1023 The resulting dataset after the data augmentation process contained a total of 18703 samples with 14419 train samples, 2142 validation samples and 2142 test samples, with 2190 nodes on average 1024 and an 8.8% of essential reactions on average. This dataset is particularly interesting as it contains a 1025 large number of graphs, which is a feature more common in graph-prediction tasks, but on average, it

also contains a large number of nodes for node-prediction. In addition, the node label depends on the global structure of the graph.

1029 1030

F EXPERIMENTAL SETUP

In this section, we describe the experimental setup used to train and evaluate the models. In addition to the architectural decision described in Section 4, we also used jumping knowledge Xu et al. (2018b) with *max* aggregation and skip connections with GNNs (ommited in Figure 5). In the case of GAT, we used 4 attention heads. Edge features were included in all GNN models except for GCN. All the models were trained with the AdamW optimizer Loshchilov & Hutter (2017).

The training was performed using NVIDIA A10 and NVIDIA Geforce RTX 3090 GPUs, both with
24 GB RAM. Given the size of the data samples, the maximum batch size that could accommodate
the GPU memory was 3 samples. The number of epochs was set to a limit of 100 epochs with 10
warmup epochs. The training of a single graph transformer model took around 24 hours.

Hyperparameters Given the computational cost of training the models, we performed a limited hyperparameter search. Generally, we optimized each of the hyperparameters in a greedy fashion, this is, we fixed the rest of the hyperparameters and optimized one at a time. Table 7 shows the main hyperparameters used for training the models.

1046	Model	Hyperparameter	Value
1047		Layers	6
1048		FNN layers	2
1049	Dese	Hidden dim	64
1050	Base	Heads	4
1051		Dropout	0.0
1052		GNN Aggr	Sum
1053		Norm	Layer Norm
1054	Ontimizer	Learning rate	0.0008
1055	Optimizer	Weight decay	1×10^{-5}
1056	Lanlagian	Freq.	10 minimum eigv.
1050	Encoder	Use attention	Yes
1057	Elicoder	Use attention	(layers = 1, heads = 2)
1058		FFN layers	2
1059	Magnetic	Freq.	10 minimum eigy.
1060	Laplacian	Use attention	No
1061	Encoder	FFN layers	2
1062		•	

Table 7: Hyperparameters used for graph transformer models.

For the ablation studies presented in Section D, in the case of the model without message passing, we used the hyperparameters in Table 8. In the case of graph transformers with undirected GNNs, we used the same hyperparameters as in Table 7. Finally, in the case of the model without transformers, we used the hyperparameters in Table 9. For the latter, we used 350 epochs instead of 100.

The FlowGAT model was trained using the same code and hyperparameters provided in the original paper Hasibi et al. (2024). The metabolic trasnformer was finetuned for the *E.coli* model using the above hyperparameters, learning rate of 0.0001 and freezing the transformer layers, except for the two prediction heads.

1063 1064

Loss function Since we are training a binary classification model, we used binary cross-entropy as loss function. Through this work, we have dealt with the prediction of reaction essentiality in a bipartite graph. Clearly, the loss $\mathcal{L}(y, \hat{y})$ was computed only for reaction nodes and not for metabolites nodes. This is:

1078

$$\mathcal{L}(y, \hat{y}) = -\sum_{i \in \mathcal{R}} y_i \log \hat{y}_i + (1 - y_i) \log(1 - \hat{y}_i)$$
(13)

¹⁰⁷³

1080	Model	Hyperparameter	Value
1081		Layers	6
1082		FNN layers	2
1083	Pasa	Hidden dim	96
1084	Dase	Heads	4
1085		Dropout	0.0
1086		Norm	Layer Norm
1087	Ontimizer	Learning rate	0.001
1088	Optimizer	Weight decay	1×10^{-5}
1089	Magnetic	Freq.	10 minimum eigv.
1000	Laplacian	Use attention	No
1001	Encoder	FFN layers	2
1091			
1092	Table 8: Hy	perparameters used f	or transformer models.
1093			
1094	Model	Hyperparameter	Value
1095		Layers	6
1096		Hidden dim	96
1097	Base	Dropout	0.0
1098	Dase	GNN Aggr	Sum
1099		Norm	Layer Norm
1100	Ontimizer	Learning rate	0.0008
1101	Optimizer	Weight decay	0.0
1102	Lanlacian	Freq.	10 minimum eigv.
1103	Encoder	Use attention	Yes
1104	Liteoder	Use attention	(layers = 1, heads = 2)
1105		FFN layers	2
1105	Magnetic	Freq.	10 minimum eigv.
1107	Laplacian	Use attention	No
1107	Encoder	FFN layers	2
1108			

1109

1110 1111 Table 9: Hyperparameters used for Dir-GNN models.

where y_i is the ground truth label for reaction *i* and \hat{y}_i is the predicted label for reaction *i*. Since the dataset is highly imbalanced, we used the weighted version of the loss function, where the weight of the positive class is the inverse of the proportion of positive samples in the dataset.

1115 1116

1117

G METABOLIC NETWORK GRAPH CONSTRUCTION

1118 In this Section, we explain how the metabolic network is transformed into a graph and how node 1119 features are extracted. The general pipeline for the transformation is shown in Figure 10. We will 1120 use this figure to exemplify the process and the design decisions taken. We start from the metabolic 1121 network shown on the left of the figure. This network is composed of 3 reactions and 2 metabolites. 1122 The first challenge that arises is the fact that certain metabolic reactions are considered reversible. 1123 This means that these reactions can be performed in both directions. We therefore need to model 1124 this fact. One naive solution is to simply link all input and output metabolites of a reaction with undirected edges. However, this solution removes the 2 sets of consumed and produced metabolites. 1125 Another solution is to label the edges of each side of the reaction (e.g. use a one-hot encoding on the 1126 edges of one side of the reaction [0, 1] and the opposite encoding on the other side [1, 0]). However, 1127 this unintentionally creates two separate sets of metabolites which might not be desirable. To solve 1128 this problem, we resorted to the approach used with LPPs when dealing with metabolic networks 1129 Ebrahim et al. (2013), and created two separate reactions for each reversible reaction. For instance, if 1130 we assume that reaction r_2 is reversible, we create two reactions with opposite direction r_2^{\leftarrow} and r_2^{\rightarrow} , 1131 as shown in the first step of the figure. 1132

1133 Once the reversible reactions are resolved, to compute the positional encodings we need to transform the bipartite graph into a unipartite graph. To achieve this, we just use a one-hot encoding to



Figure 10: Metabolic network transformation. The Figure shows the steps undertaken to transform a metabolic network into a graph and how the spectral features are computed.

differentiate between reactions and metabolites and directly treat the graph as unipartite. This is
 shown in the third graph of the figure. On this unipartite graph, we are now able to compute the
 eigenvalues and eigenvectors of the magnetic Laplacian as explained in the paper.

In the cases when it is desired to also compute the combinatorial Laplacian of the graph, we simply ignore the direction of the edges and treat the graph as undirected. This results in the fourth graph of the figure.

In addition to the positional encodings and the node features aforementioned, to provide more information to the model, we also include a flag in each node features indicating whether the node is the objective reaction, and another flag indicating whether the node is a source or sink reaction in the network.

With the procedure exposed until now, empirically it usually happened that the two reactions created from a reversible reaction had different labels. This is, if r_2 is reversible, then r_2^{\leftarrow} and r_2^{\rightarrow} could be classified differently. This is an issue since reactions r_2^{\leftarrow} and r_2^{\rightarrow} are the same reaction, just with different directions. To solve this problem, we decided to add an extra edge between the two reactions created from a reversible reaction. This helps to propagate the information during message passing and seems to solve the problem. This is shown in the fifth graph of the figure. To differentiate between the previous edges of the network, and the new edges created, we used a one-hot encoding and included it as edge features.

1169 1170 1171

1147

1148

1149 1150

H PERFORMANCE COMPARISON

Our proposed approach enables the use of GPUs to identify essential reactions in GEMs, which was not possible before. To show this, we used the model MODEL1011090001 from the Biomodels database Malik-Sheriff et al. (2020) which contains 3393 reactions and 2572 metabolites. The wall clock time, using the GPS + LapPE + Dir-Gated model described in Table 1, was of 0.09248 seconds with a GPU NVIDIA 3090 RTX with 24GB of memory. The wall clock time solving the FBA LPPs with GLPK solver was 6.7725 seconds with CPU Intel Core i5-9300H CPU @ 2.40GHz x 8

- 1179
- 1180
- 1181
- 1182
- 1183 1184
- 1185
- 1186
- 1187